Author's response to reviews

Title: The design of the SAFE or SORRY? study: a cluster randomised trial on the development and testing of an evidence based inpatient safety program for the prevention of adverse events.

Authors:

Betsie G.I. van Gaal (B.vangaal@iq.umcn.nl)
Lisette Schoonhoven (L.Schoonhoven@iq.umcn.nl)
Marlies E.J.L. Hulscher (M.Hulscher@iq.umcn.nl)
Joke A.J. Mintjes (Joke.Mintjes@han.nl)
George F. Borm (G.Borm@ebh.umcn.nl)
Raymond T.C.M. Koopmans (R.Koopmans@vphg.umcn.nl)
Theo van Achterberg (T.vanAchterberg@iq.umcn.nl)

Version: 4 Date: 23 March 2009

Author's response to reviews:

IQ healthcare
Director Prof.dr. R. Grol
Internal postal code 114
P.O. Box 9101
6500 HB Nijmegen
The Netherlands
UMC St Radboud West, route 114
Geert Grootplein 21
T +31 24 361 93 26 (day)
T + 31 6 50924855 (day and evening)
F +31 24 354 01 66
www.iqhealthcare.nl
b.vangaal@iq.umcn.nl
Section Nursing Science
Head Prof.dr. T. van Achterberg

BioMed Central Ltd
Floor 6
Dear Sir/Madam,

Enclosed please find our revised article entitled: ‘The design of the SAFE or SORRY? study: a cluster randomised trial on the development and testing of an evidence based inpatient safety program for the prevention of adverse events’, by L Schoonhoven, M Hulscher, J Mintjes, G Borm, R Koopmans, T van Achterberg and B van Gaal.

Thank you for giving the opportunity to change small typographical errors. We have found a few small errors in the article and would like to change them.

We have made a point-by-point description of the changes made in the article:

- Page 1: author Marlies Hulscher: is changed into Marlies E.J.L. Hulscher (text is highlighted)
- Page 1: author MH is changed into MEJLH (text is highlighted)
- Page 5: author BvG is changed into BGlvG (text is highlighted)
- Page 5: author MH is changed into MEJLH (text is highlighted)
- Page 7: ‘present PU’: removed an extra space between present - PU (text is highlighted)
- Page 9: author JM is changed into JAJM (text is highlighted)
- Page 9: author RK is changed into RTCMK (text is highlighted)
- Page 12: a intra correlation coefficient: changed into an intra correlation coefficient (text is highlighted)
- Page 12: author BvG is changed into BGlvG (text is highlighted)
- Page 12: author MH is changed into MEJLH (text is highlighted)
- Page 12 author JM is changed into JAJM (text is highlighted)
- Page 12 author RK is changed into RTCMK (text is highlighted)

Looking forward to hearing from you.

Yours faithfully,

On behalf of all co-authors,

Drs. Betsie van Gaal